Author's response to reviews

Title: Empirical treatment for tuberculosis in HIV: lessons from a cohort study of people living with HIV treated in Recife, Brazil

Authors:

Maria de Fatima PM Albuquerque (militaofatima@gmail.com)
Isabella Coimbra (isabella.coimbra@uol.com.br)
Joanna DL Batista (joannalyra@gmail.com)
Magda Maruza (magdamaruza@yahoo.com.br)
Ricardo AA Ximenes (ricardo.ximenes@pq.cnpg.br)
Heloisa L Ramos (helramos@terra.com.br)
Democrito B Miranda Filho (demofilho@uol.com.br)
Marcela L Santos (santosmlopes@gmail.com)
Laura C Rodrigues (laura.rodrigues@lshtm.ac.uk)

Version: 4 Date: 25 January 2014

Author's response to reviews:

Find the answers to the three reviewers attached in "Additional files Material" on upload section.